Author's response to reviews

Title: An exploration of lifestyle beliefs and lifestyle behaviour following stroke: findings from a focus group study of patients and family members.

Authors:

Maggie Lawrence (margaret.lawrence@gcu.ac.uk)
Susan Kerr (s.m.kerr@gcu.ac.uk)
Hazel Watson (h.e.watson@gcu.ac.uk)
Gillian Paton (gillian.paton@rah.scot.nhs.uk)
Graham Ellis (g.ellis@clinmed.gla.ac.uk)

Version: 3 Date: 7 December 2010

Author's response to reviews: see over
Dear Mr Cochrane

MS: 1781491336416536
An exploration of lifestyle beliefs and lifestyle behaviour following stroke: findings from a focus group study of patients and family members.
Maggie Lawrence, Susan Kerr, Hazel Watson, Gillian Paton and Graham Ellis

We have made some minor changes to the proof-reading version of the paper this include typos in the body of the text and in the reference list. A duplicate reference was found i.e. numbers 5 & 34; 34 was deleted in the reference list and reference numbering has been altered accordingly throughout the remaining document. A few words of explanation have been inserted in places in the verbatim quotes and Scottish dialect words have been anglicized.

We have addressed the issues itemised in your email re the presentation of tables etc and we have changed the manuscript title in the submission system so that it now is the same as on the manuscript.

I hope you will find these changes acceptable.

Yours sincerely

Maggie Lawrence (on behalf of the authors)